\documentclass[11pt]{article}
\usepackage{amsfonts}
\usepackage{amsthm}
\usepackage{amsmath}
\usepackage{latexsym}
\usepackage[pdftex]{graphicx}
\usepackage{url}
\usepackage{mathrsfs}
\usepackage{multicol}
\usepackage{hhline}
\usepackage{hyperref}
\hypersetup{colorlinks=true}
\setlength{\oddsidemargin}{.25in}
\setlength{\evensidemargin}{.25in}
\setlength{\textwidth}{6in}
\setlength{\topmargin}{-0.4in}
\setlength{\textheight}{8.5in}

\input{preamble.tex}

\begin{document}


\homework{1: Knot Theory}{Due Tuesday, Week 2, at the start of class}
\begin{enumerate}
\item Here are fifteen knots:
\begin{center}
\includegraphics[width=5in]{knot_table.pdf}
\end{center} 
For as many knots as you can get to, determine whether or not they are tricolorable.  If a knot is tricolorable, give me a three-coloring of that knot!  If a knot is not tricolorable, explain why it is not tricolorable.  Work on knots for at least 90 minutes or until you classify them all.
\end{enumerate}
\end{document}